Author’s response to reviews

Title: Heterotopic gastrointestinal cyst mimicking chronic cholecystitis: a case report

Authors:

Angel I Popkharitov (popkharitov@abv.bg)
Maya V Gulubova (mgulubova@hotmail.com)
Angel D Dandov (angel_dandov@mail.bg)
Dimitar P Sivrev (dsivrev@yahoo.com)

Version: 9 Date: 21 May 2008

Author's response to reviews:

MEDICAL FACULTY, THRACIAN UNIVERSITY,
Department of Surgery
Tel: (+359) 42 664 450, Fax: (+359) 42 600 710
E-mail: popkharitov@abv.bg
21.05.2008, Stara Zagora

To: The Editor – in - Chief
The Journal of Medical Case Reports
Prof. Michael Kidd,

Dear Sir,

Enclosed, please, find our revised manuscript entitled “Heterotopic gastrointestinal cyst mimicking chronic cholecystitis: a case report”.

We made the following changes to the format of our manuscript according to your advices:

1. The list of authors in the manuscript was corrected as they are in the submission system.
2. We included e-mail address for each author on the title page.
3. We removed the glossary from the manuscript text and incorporate the meaning of the word Anlage to its first citation place.

Yours sincerely,
Angel Popkharitov